```
Mauna Loa Solar Observatory Observer's Log
 ______
       Fri Jun 17 16:33:07 GMT 2011
Year: 11 Doy: 168
Observer: berkey
WEATHER COMMENT: Fri Jun 17 16:36:37 GMT 2011
Temp 48F, wind 25mph from se, wispy cirus covering the sky.
  end
**COMP PROBLEM COMMENT BY BEN**: Fri Jun 17 16:37:13 GMT 2011
COMP PC was stuck on the bios screen this morning, simply resetting didnt help.
end
Fri Jun 17 16:43:46 GMT 2011: PSPT Start Patrol
Fri Jun 17 16:46:41 GMT 2011 H-ALPHA Start Patrol
Fri Jun 17 16:47:00 GMT 2011
                               CHIP
                                        Start Patrol
COMP COMMENT BY BEN: Fri Jun 17 16:59:01 GMT 2011
I got the CoMP computer after removing the BIOS battery and leaving it unplugged for about 10 minutes. This looks like th
e problem we were seeing alot before replacing the powersupply a couple months ago.
  end
Fri Jun 17 18:02:43 GMT 2011
                               CHIP
Fri Jun 17 18:04:17 GMT 2011
                               CHIP
                                        End LSD
Fri Jun 17 18:04:23 GMT 2011
                               CHIP
                                        BiasLSD
Fri Jun 17 18:05:01 GMT 2011
                               CHIP
                                        End BiasLSD
Fri Jun 17 18:05:07 GMT 2011
                               CHIP
                                        Bias
Fri Jun 17 18:05:39 GMT 2011
                               CHIP
                                        End Bias
Fri Jun 17 18:05:44 GMT 2011
                               CHIP
                                       ReStart Patrol
Fri Jun 17 19:43:14 GMT 2011
                               CHIP
                                        End Patrol
Fri Jun 17 19:43:38 GMT 2011: PSPT Start Patrol
Fri Jun 17 19:44:51 GMT 2011 H-ALPHA End Patrol
Fri Jun 17 19:52:36 GMT 2011 H-ALPHA Start Patrol
Fri Jun 17 19:52:47 GMT 2011
                               CHIP
                                        Start Patrol
Fri Jun 17 22:50:04 GMT 2011: PSPT Start Patrol
Fri Jun 17 23:01:44 GMT 2011
                               CHIP
                                        LSD
Fri Jun 17 23:03:29 GMT 2011
                               CHIP
                                        End LSD
Fri Jun 17 23:03:35 GMT 2011
                               CHIP
                                        BiasLSD
Fri Jun 17 23:04:09 GMT 2011
                               CHIP
                                        End BiasLSD
Fri Jun 17 23:04:15 GMT 2011
                                        Bias
                               CHIP
Fri Jun 17 23:04:45 GMT 2011
                               CHIP
                                        End Bias
Fri Jun 17 23:04:50 GMT 2011
                               CHIP
                                       ReStart Patrol
Sat Jun 18 01:03:26 GMT 2011
                               CHIP
                                        End Patrol
Sat Jun 18 01:04:14 GMT 2011: PSPT Abort Patrol
Sat Jun 18 01:05:51 GMT 2011 H-ALPHA End Patrol
GONG COMMENT BY BEN: Sat Jun 18 01:09:47 GMT 2011
Replaced gong Tape drives today. Bert reported they were giving some wierd issues with the data tapes.
```

NOTE BY BEN: Sat Jun 18 01:10:28 GMT 2011

Winds seem to be picking up and the cirus is getting thicker, I am headed to fedex to mail out the tape drives. for gong.

____end___

Sat Jun 18 01:11:06 GMT 2011 MkIV